Author's response to reviews

Title: TENS Attenuates CFA-induced Hyperalgesia and Inhibits Spinal ERK1/2-COX-2 Pathway Activation in Rats

Authors:

Junfan Fang (fangjunfan0223@163.com)
Yi Liang (liangyiwww@126.com)
Junying Du (dujunying0706@163.com)
Jianqiao Fang (fangjianqiao7532@163.com)

Version: 2 Date: 4 January 2013

Author's response to reviews: see over
Dear Editor-in-Chief,

Please find our manuscript entitled " TENS Attenuates CFA-induced Hyperalgesia and Inhibits Spinal ERK1/2-COX-2 Pathway Activation in Rats ", which we would like to submit for publication as basic research papers in *BMC Complementary & Alternative Medicine*.

We confirm that this manuscript has not been published elsewhere and are not currently under consideration by another journal published. The study complies with current ethical consideration. All authors have approved the manuscript and agree with article's publication in BMC Complementary & Alternative Medicine. The study was supported by the National Natural Science Fund of China, Zhejiang Provicinal Natural Science Foundation of China, Research Fund of Zhejiang First-foremost Key Subject--Acupuncture & Moxibustion. The authors have no conflicts of interest to declare.

Please address all correspondence to:

Professor Jian-Qiao Fang, MD, PhD, Department of Neurobiology and Acupuncture Research, Zhejiang Chinese Medical University, 548 Binwen Road, Binjiang District, Hangzhou City, Zhejiang Province 310053,China. Tel:(+86) 571-8667-3000, Fax: (+86) 571-8667-3000, E-mail: fangjianqiao7532@163.com
We appreciate your consideration of our manuscript, and we look forward to hearing from you at your earliest convenience.

Sincerely yours,

Jian-Qiao Fang, MD, PhD.
Professor of Acupuncture & Physiology
Vice-president of Zhejiang Chinese Medical University